Author's response to reviews

Title: Postural changes in women with chronic pelvic pain: a case control study

Authors:

Mary ML Montenegro (montenegro@usp.br)
Elaine EL Mateus- Vasconcelos (elainelemes@baraodemaua.br)
Julio JC Rosa e Silva (juliocrs@convex.com.br)
Francisco FJ Candido dos Reis (fjcreis@fmrp.usp.br)
Antonio AN Nogueira (aanoguei@fmrp.usp.br)
Omero OP Poli Neto (polineto@fmrp.usp.br)

Version: 4 Date: 16 March 2009

Author's response to reviews: see over
Cover Letter

Dear Editor BMC Musculoskeletal Disorders

We are pleased to submit the revised version of our manuscript “Postural assessment of women with chronic pelvic pain: a case control study” for consideration for publication in BMC Musculoskeletal Disorders.

We have now accepted the suggestions made by the reviewers. We thank you and the reviewers for improving the quality of this manuscript.

The manuscript has neither been published nor is currently under consideration for publication either in whole or in part by any other journal. The submission has been read and approved by each of my co-authors.

Thank you very much.

Prof. Omero Benedicto Poli Neto, MD, PhD
University of Sao Paulo
Faculty of Medicine of Ribeirao Preto
Department of Obstetrics and gynecology
Av. Bandeirantes, 3900
Ribeirao Preto - SP - Brasil
14049-900

Email: polineto@fmrp.usp.br
Phone: +55-16-36022961
Fax: +551636330946
REPLIES TO REVIEWER #1:

Comments to authors:
Reviewer #1: Abstract: Replace “We therefore determined the frequency of postural changes in women with CPP2 with- We aimed to determine if the frequency of postural changes was higher in women with CPP than healthy subjects.

Reply:
We thank you for the kind comments on our study and for helping to improve the quality of our manuscript. We have now made the following modifications in accordance with your suggestions. Please see page 2, line 34.

Conclusion: We have shown here that it is possible to determine the frequency of postural changes in women with CPP using only clinical examination - it should read- From this study we conclude that postural changes are seen more frequently in women with CPP

Reply:
Change was made. Please see page 7, line 162.